Author's response to reviews

Title: Clinical Characteristics and Outcomes of Patients with Acute Myelogenous Leukemia Admitted to Intensive Care: A Case-Control Study

Authors:

Amanda Roze des Ordons (alr@ualberta.ca)
Kris Chan (krisc@ualberta.ca)
Imran Mirza (Imran.Mirza@albertahealthservices.ca)
Derek R Townsend (drt1@ualberta.ca)
Sean M Bagshaw (bagshaw@ualberta.ca)

Version: 2 Date: 9 March 2010

Author's response to reviews: see over
Re: Manuscript Submission titled “Clinical Characteristics and Outcomes of Patients with Acute Myelogenous Leukemia (AML) Admitted to Intensive Care: A Case-Control Study”

Please find enclosed a manuscript for consideration for publication in BMC Cancer as an original article.

This study is a retrospective case-control study of patients with acute myelogenous leukemia (AML) requiring life-sustaining therapy in the intensive care unit. This is the first study to examine incidence estimates of AML patients admitted to ICU along with inclusion of case-matched hospitalized AML and non-AML ICU controls. We believe this study will be of interest to your readership.

The manuscript has been reviewed and approved by all authors. The manuscript has not been previously published and is not being considered for publication elsewhere.

If there are any further questions or concerns, please contact us at your convenience. We look forward to your review.

Sincerely,

Sean M Bagshaw
Amanda L. Roze des Ordons
Kris Chan
Imran Mirza
Derek R. Townsend

Sean M Bagshaw, MD, MSc, FRCPC
Division of Critical Care Medicine,
University of Alberta Hospital,
3C1.12 Walter C. Mackenzie Centre, 8440-122 Street
Edmonton, Alberta T6G2B7 Canada
Telephone: (780) 407-6755
Fax: (780) 407-1228
Email: bagshaw@ualberta.ca